Author's response to reviews

**Title:** Automated detection of regions of interest for tissue microarray experiments: An image texture analysis

**Authors:**

Bilge Karacali (bilge@drexel.edu)
Aydin Tozeren (aydin.tozeren@drexel.edu)

**Version:** 8  **Date:** 7 March 2007

**Author's response to reviews:** see over
Dear Editor,

Please find attached a revised version of the article entitled “Automated Detection of Regions of Interest for Tissue Microarray Experiments: An Image Texture Analysis” by B. Karacali and A. Tozeren (5134365331133852) with the formatting changes requested by the editorial office.

Also included with this cover letter is the list of changes made in the manuscript in the proofing stage. We hope that you will find the revised manuscript appropriate for publication in the journal and look forward to hearing from you soon.

Sincerely yours,

Bilge Karacali
List of changes

We have removed the authors' qualifications from the manuscript and submissions system.

We have reformatted the author affiliations as requested.

We have removed the horizontal line from above the corresponding author indication.

We have indicated the reference to Figure 10 in proper numerical order.

We have included a Discussion section between the Results and Conclusions sections.

We have corrected the section title summarizing authors' contributions, and included a sentence stating that both authors read and approved the final manuscript.

We have changed the title of the figures section to “Figure legends”.

We have removed the additional file from the submission system.